Author's response to reviews

Title: Health-Economic evaluation of vaccination strategies for the prevention of herpes zoster and postherpetic neuralgia in Germany

Authors:

Bernhard Ultsch (UltschB@rki.de)
Felix Weidemann (WeidemannF@rki.de)
Thomas Reinhold (Thomas.Reinhold@charite.de)
Anette Siedler (SiedlerA@rki.de)
Gérard Krause (KrauseG@rki.de)
Ole Wichmann (WichmannO@rki.de)

Version: 2 Date: 14 July 2013

Author's response to reviews: see over